# Effects of Family-centered Interventions on Self-efficacy of Families of Patients With Low Consciousness Admitted to the Intensive Care Units: A Randomized Controlled Trial

Hadis Biranvand¹00, Mehdi Birjandi²00, Fateme Goudarzi³00, Heshmatolah Heydari4, ⁵\*00

1. Nursing (MSc), School of Nursing and Midwifery, Lorestan University of Medical Sciences, Khorramabad, Iran.

2. Assistant Professor, Department of Biostatistics and Epidemiology, School of Health and Nutrition, Lorestan University of Medical Sciences, Khorramabad, Iran. 3. Assistant Professor, Department of Nursing, School of Nursing and Midwifery, Lorestan University of Medical Sciences, Khorramabad, Iran.

4. Associate Professor, Department of Community Health Nursing, Social Determinants of Health Research Center, School of Nursing and midwifery, Lorestan University of Medical Sciences, Khorramabad, Iran.

5. French Institute of Research and High Education (IFRES-INT), Paris, France.



**Citation** Biranvand H, Birjandi M, Goudarzi F, Heydari H. Effects of Family-Centered Interventions on Self-efficacy of Families of Patients With Low Consciousness Admitted to the Intensive Care Units: A Randomized Controlled Trial. J Holist Nurs Midwifery. 2023; 33(1):69-77. https://doi.org/10.32598/jhnm.33.1.2401

Running Title Effect of Family-centered Interventions on Self-efficacy of Family of Patients with Low Consciousness

doi https:/doi.org/10.32598/jhnm.33.1.2401



Article info: Received: 23/02/2022 Accepted: 4/10/2022 Available Online: 01/01/2023

#### Keywords:

Family nursing, Self-efficacy, Consciousness disorders, Intensive care units

## ABSTRACT

**Introduction:** Families of patients with low consciousness have an important role in supporting them and should acquire the necessary capability for patient care.

**Objective:** This study aimed to examine the effect of family-centered interventions on the self-efficacy of the families of patients admitted to the intensive care units.

**Materials and Methods:** This research was a randomized clinical trial with a pretest-posttest design conducted in intensive care units of hospitals in Khorram abad City, Iran, in 2019-2020. Seventy family members of patients were selected by convenience sampling and were assigned to the intervention (n=28) and control (n=28) groups by stratified random blocks. The study data were collected through the self-efficacy questionnaire and Hospital Anxiety and Depression Scale. The intervention was carried out using a family-centered model. The collected data were analyzed using the statistical tests of the Chi-square, independent t-test, and paired t-test. Also, the generalized linear model was used to compare changes in self-efficacy scores of the two study groups after adjusting demographic, anxiety, and depression variables.

**Results:** Data analysis showed that most samples were male (53.8%) with a mean age of 38.15±8.68 years. The results showed a significant difference in the mean scores of self-efficacy, anxiety, and depression between the intervention and control groups (P=0.001). The intervention significantly increased the Mean±SD of caregivers' self-efficacy (40.14±14.35, P=0.001) in the intervention group. The results showed that the self-efficacy mean score of the intervention group was 36 scores higher than the caregivers' self-efficacy of the control group (95% CI; 20.6-51.4, P=0.028).

**Conclusion:** The participation of family members of patients with low consciousness in the care of the patients in the intensive care unit has an effective role in the self-efficacy of caregivers of these patients.

# \* Corresponding Author:

Heshmatolah Heydari, Associate Professor.

Address: Social Determinates of Health Research Center, Lorestan University of Medical Sciences, Khorramabad, Iran. Tel: +98 (66) 33120140 E-mail: h-hidari@razi.tums.ac.ir



Highlights

- A patient's admission to the intensive care unit and recovery process are challenges for the patient's family.
- Participation of family members in patient care could positively affect the recovery process.

• Self-efficacy is an individual perception of abilities for creating the desired outcome, which can organize cognitive, social, emotional, and behavioral skills for achieving various goals efficiently.

• Family-centered interventions could improve self-efficacy in patients' families and help healthcare providers make effective decisions for their patients.

## **Plain Language Summary**

Families of patients with low consciousness have an important role in supporting them and should acquire the necessary capabilities for patient care. This study was conducted using by randomized clinical trial methodology. Seventy family members of patients were assigned to the intervention and control groups. Data were collected to evaluate the self-efficacy, anxiety, and depression of patients' families. Results revealed that family-centered training could improve the self-efficacy, depression, and anxiety of patients' families hospitalized in intensive care units. It is suggested that hospital authorities carry out these programs based on the guidelines for patients' families from admission to the intensive care unit, during hospital stays, until discharge.

### Introduction

he Intensive Care Unit (ICU), as one of the subspecialty units in hospitals, has particular conditions and equipment to which patients with critical conditions are admitted [1]. Most patients in the ICU have varying levels of consciousness. Low consciousness and coma result from traumatic and non-traumatic causes, including poisoning, metabolic disorders, infections, head injuries, and primary central nervous system lesions. Metabolic disorders and poisoning are considered the most important causes of low consciousness [2, 3]. Admission of a patient to the intensive care unit and the recovery process is challenging for the patient's family, resulting from a lack of knowledge about diagnosis, prognosis, and treatment [4]. Patients' families confront unfamiliar environments, therapeutic procedures, and unclear patient status, so they experience varying levels of fear and pressure over confronting the admission of a family member to the ICU [5, 6]. Many studies have shown that admission of a family member to the ICU can affect the mental and psychological health of other family members and result in disorders such as depression [7], anxiety, fear, and sleep complaints [8].

Patients' families are involved during admission, hospitalization, discharge, and continuing home care [9, 10]. Patients with low consciousness cannot make decisions, so healthcare providers should reach a decision

with family members, or a family member should be the attorney of decision-making [11]. Also, evidence shows that the participation of family members in patient care could positively affect the recovery process [9, 12]. This outcome requires that family members identify the situation and acquire essential self-efficacy for managing this condition. Self-efficacy is a construct that emerged in the social cognitive theory [13] and is defined as an individual perception of abilities for creating desired outcomes. Self-efficacy can efficiently organize cognitive, social, emotional, and behavioral skills for achieving various goals [14]. Many factors, such as anxiety or depression, affect self-efficacy [15]. According to studies, patients' families admitted to the ICU suffer psychological distress, which can negatively affect the family members and the patient [16]. Stressors for patients' families include invasive care, complex equipment, unfamiliar interventions, cardiopulmonary resuscitation before death, and losing the loved one in these units [17, 18]. These families continuously try to adapt to critical conditions in the intensive care unit [19].

On the other hand, the patients' families should manage the situation to make proper decisions together with healthcare providers [20]. Studies have shown that family participation in patient care positively affects their recovery process. This participation requires that patients' families acquire essential empowerment with regard to the patient's care. A study shows that patients' families in the ICU need various physical and psychological support [4]. Another study shows that providing information about patient status, treatment plans, and psychological support are the most important needs of families [21]. The Critical Care Association has introduced a family-centered care model to involve the family in caring for patients with low consciousness. This model involves attending family members in the ICU, providing family support, communicating with family members, empowering the family for patient care, considering the appropriate environment for attending the hospital, and special consultation for palliative care [22]. Patients' families have adequate motives to participate in patient care.

On the other hand, their patients may need family care and help for months outside the hospital and at home. No program has been developed to empower caregivers of patients with low consciousness in Iran, while a program should be developed based on the evidence before providing any healthcare. This study aimed to determine the effect of the family-centered intervention on the self-efficacy of the families of people with low consciousness in the ICUs.

## **Materials and Methods**

This randomized controlled trial was conducted on 70 family members of patients with low consciousness hospitalized in the ICUs of 2 hospitals in Khorram abad City, Iran, from November 2019 to March 2020. The inclusion criteria were as follows: being family members of patients admitted to the ICU for low consciousness, having direct contact with the patient, being primary caregiver, expected to be in the ICUs for 2 weeks, being physically and mentally healthy (based on self-expression), being literate, being willing to participate in the study, lacking an academic degree in health sciences (medicine, nursing, and midwifery), and not having experience in caring patients with low consciousness. The exclusion criteria were as follows: unavailability of cases for any reason (discharged or deceased patients), refusal to continue cooperation, lack of continuing patient care, and inappropriate physical and mental health status.

The inclusion criteria for patients were low consciousness due to traumatic and non-traumatic causes, Glasgow coma scale of 7 to 10, aged 18 to 60 years, and being at the ICU for at least 2 weeks. The exclusion criteria for patients were no access to the patient (death and transfer to the ward or other hospital) and the unwillingness of the patient's family to participate in the study. The sample size was estimated at 28 in each group considering  $\alpha$ =0.05,  $\beta$ =0.2, d=3, and SD=4. Given the dropout of 20%, the sample size was estimated at 35 subjects for each group. To determine the value of the standard deviation, considering that no similar study was available, the value of the standard deviation was calculated based on the R/6 index. The R value of the range of changes is based on the difference between the minimum and maximum score of the questionnaire, which is calculated as 21. Also, based on the mentioned index, the standard deviation value was estimated to be 4.

Sampling was carried out using convenience and sequential methods, and subjects were assigned to the intervention (n=35) and control (n=35) groups by stratified random blocks (to match the two groups in terms of sex). By considering sex as a category in men and women, the samples were located in the studied groups in random blocks of 4. After visiting the ICU and studying the patients' files and medical histories, the researcher selected the eligible subjects. Finally, the sample size was considered to be 56 because of dropouts (Figure 1).

For data gathering, various tools, such as questionnaires for assessing demographic information, self-efficacy, and the Hospital Anxiety and Depression Scale (HADS), were used. In the demographic section, variables such as age, gender, occupation, education, and type of disease (traumatic or nontraumatic) were assessed.

In this study, a self-efficacy tool was developed based on the questionnaire by Nolan et al. [23] for examining the self-efficacy of the families of people with low consciousness. This questionnaire consists of 40 questions that examine various dimensions of caring for patients with low consciousness in the ICU. The questionnaire is scored on a 5-point Likert scale from completely disagree=1 to completely agree=5. A higher score shows better self-efficacy. The questionnaire was developed based on the ability of family members in various dimensions, including suctioning; nutrition; oxygen therapy; care of skin, mouth, and eye; and patient hygiene. The total score ranges from 40-200. A low score indicates low self-efficacy, and a high score indicates high self-efficacy of subjects in patient care. The level of self-efficacy was categorized into low (40-94 score), moderate (95-148 score), and high (149-200 score). The quantitative and qualitative methods were used to determine the content validity. In the qualitative phase, 10 healthcare providers (2 physicians, 3 clinical nurses, and 5 nursing educators) in the field of critical care assessed the content validity. In the quantitative stage, the content validity ratio (CVR) and the content validity index

(CVI) were calculated, which were found acceptable. The reliability of the questionnaire was obtained using the Cronbach  $\alpha$  as 0.98.

In this study, anxiety, and depression were examined as moderators and assessed using the HADS questionnaire [24]. Researchers used the psychometric version of this questionnaire in the Persian language [25]. This questionnaire has two subscales of anxiety and depression, with 7 questions for each subscale. Each question is scored on a 4-point Likert scale (from 0 to 4); the respondents respond based on their feelings. A score of 3 indicates maximum anxiety and depression, and a score of 0 indicates minimum anxiety and depression. The total score of depression and anxiety scales ranges from 0 to 21. Score 14–21 indicates severe anxiety and depression, 7–14 moderate disorders, and 0–7 health status.

The study intervention was implemented based on a family-centered model [22]. Before the study, self-efficacy, anxiety, and depression questionnaires were provided for eligible caregivers to complete. The intervention was conducted in three steps for each family (Table 1). One week after all interventions, questionnaires of self-efficacy, anxiety, and depression were provided for caregivers to complete by the intervention and control groups. Considering that it was not possible to meet patients in the ICU, the researcher called the intervention group for training; thus, the control and intervention groups did not have contact with each other. In this process, the control group received routine care from healthcare providers in the ICU wards. For ethical considerations, necessary teaching was provided to the control group after completing the study.

Data analysis was carried out using descriptive and inferential statistics in SPSS v. 19. Statistical tests of the independent t-test, paired t-test, and Chi-square were used for data analysis. Also, the generalized linear model (GLM) was used to compare changes in the self-efficacy scores of the two study groups, adjusting demographic, anxiety, and depression variables. The results were reported at a significance level of 0.05.

## Results

Data analysis showed that most subjects were male (53.8%) with a mean age of  $38.15\pm8.68$  years (Table 2). The results showed that in the intervention group, the mean self-efficacy score of subjects before the intervention (137.78±13.81) increased compared to after the intervention score (177.92±12.36), and this difference (40.14±14.35) was significant (P=0.001). The

results showed that in the control group, the mean self-efficacy score of samples before the intervention (135.78 $\pm$ 16.66) decreased compared to after the intervention (133.25 $\pm$ 21.57), and this difference (2.53 $\pm$ 26.57) was not significant. The results also showed a significant difference (P=0.001) in the means of self-efficacy score between the control (133.25 $\pm$ 21.57) and intervention (177.92 $\pm$ 12.36) groups, so the intervention increased the mean self-efficacy score significantly (Table 3).

The results showed that in the intervention group, the Mean±SD anxiety score of subjects before the intervention (15.42±3.84) decreased compared to after the intervention score (9.78±2.79), and this difference (5.64±3.65) was significant (P=0.001). The results showed that in the control group, the mean anxiety score of subjects before the intervention (12.53±4.85) increased compared to after the intervention score (13.21±3.31), and this difference (0.68±5.65) was not significant (P=0.53). Results of this study showed a significant difference in the Mean±SD of anxiety score between the control (13.21±3.31) and intervention (9.78±2.79) groups, so intervention decreased the mean anxiety score significantly (P=0.001). The results also showed that in the intervention group, the Mean±SD depression score of subjects before the intervention (14.53±5.3) decreased compared to after the intervention score (9.07±3.04), and this difference (5.42±5.48) was statistically significant (P=0.001). The results showed that in the control group, the Mean±SD depression score of subjects before the intervention (11.39±5.16) increased compared to after the intervention score (12.25±4.35), and this difference (0.85±6.24) was not significant. The results also showed a significant difference (P=0.001) in the mean depression scores between the control (12.25±4.35) and intervention (9.07±3.04) groups, so the intervention decreased the mean of depression significantly (Table 3).

To compare changes in self-efficacy scores of the two study groups, after adjusting for demographic, anxiety, and depression variables by using the generalized linear model (GLM), the results showed that the self-efficacy mean score of the intervention group was 36 scores higher than the caregivers' self-efficacy of the control group. These differences were significant statistically (P=0.001). The results showed, assuming the fixed value of other variables by GLM, the self-efficacy of women's caregivers was 11.38 scores higher than men's caregivers. These differences were significant statistically (P=0.028). Also, caregivers' self-efficacy scores of traumatic patients were 21.91 higher than non-traumatic patients (P=0.028), and these differences were significant statistically (Table 4).

## Table 1. Steps of the intervention type for study

Sessions	Interventions
1 <sup>st</sup>	This session was carried out 24 hours after the patients' admission to the ICU. After being familiar with fam- ily members and developing trust between subjects and investigator, an explanation was provided about the patient's status, disease, possible complications, and the necessity of cooperation of family members with the healthcare team and the benefit of their communication with the patient for the patient's recovery. Care that patients need at the hospital and home was also explained, including care of skin and eyes, nutrition, urinary tract, respiratory tract, musculoskeletal tract, and prevention of bed sore for the patient. A caring booklet and pamphlet developed by the research team were also provided for family members, including content about patient care. Teaching was conducted individually in a quiet place in the resting room of the ICU staff.
2 <sup>nd</sup>	In the second step, after identifying caregivers' needs and coordinating with the head nurse of the ICU and the director of the hospital, caregivers were asked to attend three 1-hour sessions at the bedside in the evening shift, individually. Teaching was conducted over three 1-hour sessions at the bedside every other day for one week. Caregivers participated in patients' care as an assistant, and they were offered to practice. Teaching was conducted so that caregivers felt that they could participate as the assistant of the nurse in care plans. In this section, teaching was provided about hygiene, nutrition, change of position, bed sore, suctioning, etc. All theoretical and practical teaching was carried out by the investigator, who is a nurse in the ICU.
3 <sup>rd</sup>	In the third step, caregivers were asked to practice the taught content for the patient and participate in patient care. In this stage, if caregivers could not care properly or had a question regarding training and needed further teaching, the investigator taught theoretically and practically them again.

## Discussion

This study showed that family-centered intervention increased self-efficacy in family members of patients with low consciousness in the ICU. In this study which was conducted using a family-centered model, the investigator was a nurse, attended to the teaching caregivers, and provided care for the patients. Findings of another study showed that theoretical and practical teaching of informal caregivers of patients with stroke

Table 2.	Comparing	demographic	variables	between	study groups
----------	-----------	-------------	-----------	---------	--------------

Variables		No	р		
		Control group	Intervention group	r	
Sex	Male	14(50)	16(57.1)	0 700*	
	Female	14(50)	12(42.9)	0.789	
Married	Single	8(28.5)	6(21.4)	0.759*	
Marrieu	Married	20(71.4)	22(78.5)	0.758*	
	Under diploma	7(25)	4(14.2)		
Education	Diploma	9(32.1)	10(35.7)	0.685*	
Education	Associate degree	4(14.2)	3(10.7)		
	Bachelor	8(28.5)	11(39.2)		
	Self-employment	9(32.1)	11(39.2)		
doſ	Employment	5(17.8)	5(17.8)	0.716*	
	Jobless	14(50)	10(35.7)		
Cause of admission	Traumatic	6(21.4)	1(3.5)	0 102*	
	Non-traumatic	22(78.5)	27(96.4)	0.133	
Age (Mean±SD)	-	37.21±8.78	39.82±8.58	0.266**	

\* The Chi-square test. \*\* The Independent t-test.

Variables	Groups		P*	P**			
		Before Intervention	After Intervention	Mean Difference	Groups	Groups	
Self-efficacy	Control	135.78±16.66	133.25±21.57	2.53±26.57	0.62	0.001	
	Intervention	137.78±13.81	177.92±12.36	40.14±14.35	0.001	0.001	
Anxiety	Control	12.53±4.85	13.21±3.31	5.65±0.67	0.53	0.001	
	Intervention	15.42±3.84	9.78±2.79	5.64±3.65	0.001	0.001	
Depression	Control	11.39±5.16	12.25±4.35	6.24±0.85	0.474	0.001	
	Intervention	14.53±5.3	9.07±3.04	5.42±5.48	0.001	0.001	

Table 3. Comparing the mean scores of self-efficacy, anxiety, and depression between two groups, before and after the study

\*The paired t-test; \*\*The Independent t-test.

based on patients' needs and caregivers' abilities could increase their practical skills [26]. Education, along with the care provided, allows the caregiver to observe the educator during the performance. Resources of self-efficacy include live role models, confrontation with action, positive assessment, and mastery of activity. For confrontation with action and positive assessment, the caregiver acts based on the guidelines and is evaluated during the action, empowering the caregivers in their tasks.

This study showed that the depression and anxiety of caregivers improved in addition to their self-efficacy. In contrast, another study conducted on caregivers

**Table 4.** Comparing changes in the mean scores of caregivers' self-efficacy after the adjustment of demographic, anxiety, and depression variables using the generalized linear model

Variables	Groups	В	SE	95% CI Lower-Upper	Р
Crown	Control	Reference			
Group	Intervention	36	7.85	20.6-51.4	0.001
	Married	Reference			
Maritai status	Single	2.17	6.59	-10.74-15.08	0.742
6	Male	Reference			
Sex	Female	11.38	5.17	1.25-21.52	0.028
Discos	No	Reference			
Disease	Yes	21.98	8.54	5.16-38.67	0.01
Age		-0.17	0.32	-0.79-0.46	0.61
Changes of anxiety		1.72	1.1	-0.45-3.89	0.121
Changes of depression		-1.83	1.03	-3.85-0.19	0.76
	Control	Reference			
Group* Changes in anxiety	Intervention	-2.67	1.69	-5.98-0.65	0.115
	Control	Reference			
	Intervention	1.12	1.33	-1.51-3.74	0.4





Figure 1. Diagram study based of CONSORT flowchart

of patients with cancer showed that these interventions do not affect caregivers' mental and psychological status; however, they improved their self-efficacy [27]. Inappropriate psychological status results in caregivers' inappropriate self-efficacy in interaction with healthcare providers and inappropriate expectations from the ICU and its staff [28].

This study showed the effectiveness of family-centered interventions in improving families' self-efficacy. Although empowering caregivers has many benefits in improving patients' care and facilitating cooperation with staff, it should be considered that the ICU ward is a special unit that needs silence and peace. Nosocomial infections are common in these units. Patients' caregivers should attend based on proper planning, as crowded ICUs by patients' caregivers could result in increasing nosocomial infections, busy units, and burnout in caregivers of these units. In some developed countries, there are some guidelines for supporting families of patients admitted to the ICUs, which meet family needs over the admission of patients [20, 29]. It is suggested to plan for attending to patients' families considering patient safety according to the evidence.

Another element of the family-centered model is communication and increasing knowledge of patients' families. These families spend tough days; therefore, communication with them needs particular skills. Healthcare providers in the ICU should be familiar with the principles of proper communication with the family of critically-ill patients. It is required that authorities of continuing education consider the principles of communication with family and patients and teach nurses in the ICUs on how to deliver bad news to patients' families.

Another element of the family-centered model is using specialized counselors for the peace of family of critically-ill patients. Mortality and disability rates are high in patients admitted to the intensive care unit [30] and could be a source of anxiety and depression, and low self-efficacy. These families are patients' attorneys for decision-making and should be accurately informed for proper decision-making [31]. Studies have shown that clarification and understanding of patients' families can reduce hospital stays, which can result in further vacant beds for critically ill patients and decrease costs [32, 33].

In this study, teaching was carried out by the investigator, who is a nurse in the ICU and was familiar with patients' condition and their families. It is ideal for supporting patients' families comprehensively by healthcare professionals, including psychologists, social workers, and religious experts. Palliative care should be considered for the patient's family from admission so that the family can tolerate the suffering of the deceased patient and grieve easily and endure fewer psychological disorders [22].

One of the limitations of this study was its short-term intervention. It is suggested to conduct this study over a long-term period so that family caregivers receive proper teaching and support since the patient's admission to the intensive care unit. Then, the self-efficacy of caregivers and the effect of these interventions on patients' recovery are examined over time. In the family-centered model, various experts should be used in the healthcare team for intervention. In this study, the investigator, a nurse in the intensive care unit, carried out all interventions due to existing limitations. Another limitation of this study was that few studies had been conducted on this issue. It is suggested to conduct a further study to examine this topic from different angles. In this study, the investigator conducted teaching in the resting room of the staff. There was no place for sleep and rest of caregivers at hospitals, which could result in mental and psychological disorders in patients' families. It is suggested that hospital authorities consider these points in the management and provide places for patients' caregivers at the hospitals.

Family-centered interventions could improve self-efficacy in patients' families, and consequently, healthcare providers could make effective patient decisions. Therefore, it is suggested that hospital authorities carry out these programs based on the guidelines for patients' families from admission to the ICU and during their stays in the hospital until discharge. As less is known about the effect of these interventions on patients' recovery in the ICUs, it is suggested to examine the effect of these interventions on the recovery process and outcomes of patient care.

## **Ethical Considerations**

### **Compliance with ethical guidelines**

The purpose and method of the study were explained to all participants. The participants were assured of anonymity and confidentiality and the right to withdraw from the study. Informed consent was obtained from the participants. The study was approved by the Research Ethics Committee of Lorestan University of Medical Sciences (LUMS.REC.1397.196). A Code of a clinical trial was also obtained (IRCT20180721040540N2).

## Funding

This study was supported by the Deputy of Research, Lorestan University of Medical Sciences.

### Authors' contributions

Data collection: Hadis Biranvand; Data analysis: Mehdi Birjandi; Study supervision: Mehdi Birjandi, Fatemeh Goudarzi, Heshmatolah Heydari; Writing-original draft: Hadis Biranvand and Heshmatolah Heydari; Methodology and Final approval of the article: All authors.

#### **Conflict of interest**

The authors declared no conflict of interest.

#### Acknowledgments

The authors would like to thank all individuals who supported and helped them to conduct this study freely.

## References

- [1] Marshall JC, Bosco L, Adhikari NK, Connolly B, Diaz JV, Dorman T, et al. What is an intensive care unit? A report of the task force of the world federation of societies of intensive and critical care medicine. Journal of Critical Care. 2017; 37:270-6. [DOI:10.1016/j. jcrc.2016.07.015] [PMID]
- [2] Horsting MW, Franken MD, Meulenbelt J, van Klei WA, de Lange DW. The etiology and outcome of non-traumatic coma in critical care: A systematic review. BMC Anesthesiology. 2015; 15:65. [DOI:10.1186/s12871-015-0041-9] [PMID] [PMCID]
- [3] Alizadeh A, Rakhshanizadeh F. Loss of consciousness in a little traveler. Caspian Journal of Internal Medicine. 2021; 12(1):103-6.
   [DOI:10.22088/cjim.12.1.103] [PMID] [PMCID]
- [4] Khatri Chhetri I, Thulung B. Perception of nurses on needs of family members of patient admitted to critical care units of teaching hospital, Chitwan Nepal: A cross-sectional institutional based study. Nursing Research and Practice. 2018; 2018:1369164. [DOI:10.1155/2018/1369164.] [PMID] [PMCID]
- [5] Johnson CC, Suchyta MR, Darowski ES, Collar EM, Kiehl AL, Van J, et al. Psychological sequelae in family caregivers of critically iii intensive care unit patients. A systematic review. Annals of the American Thoracic Society. 2019; 16(7):894-909. [DOI:10.1513/ AnnalsATS.201808-540SR] [PMID]
- [6] Choi J, Donahoe MP, Hoffman LA. Psychological and physical health in family caregivers of intensive care unit survivors: Current knowledge and future research strategies. Journal of Korean Academy of Nursing. 2016. 46(2):159-67. [DOI:10.4040/ jkan.2016.46.2.159] [PMID]

- [7] Rusinova K, Kukal J, Simek J, Cerny V, DEPRESS study working group. Limited family members/staff communication in intensive care units in the Czech and Slovak Republics considerably increases anxiety in patients' relatives--the DEPRESS study. BMC Psychiatry. 2014; 14:21. [DOI:10.1186/1471-244X-14-21] [PMID] [PMCID]
- [8] Day A, Haj-Bakri S, Lubchansky S, Mehta S. Sleep, anxiety and fatigue in family members of patients admitted to the intensive care unit: A questionnaire study. Critical Care. 2013; 17(3):R91. [DOI:10.1186/cc12736] [PMID] [PMCID]
- [9] Adelman RD, Tmanova LL, Delgado D, Dion S, Lachs MS. Caregiver burden: A clinical review. JAMA. 2014; 311(10):1052-60. [DOI:10.1001/jama.2014.304] [PMID]
- [10] Nikbakht Nasrabadi A, Shahsavari H, Almasian M, Heydari H, Hazini A. Designing a process model of home care service delivery in Iran: A mixed methods study. International Journal of Community Based Nursing & Midwifery. 2019; 7(4):288-99. [DOI:10.30476/ ijcbnm.2019.73934.0] [PMID]
- [11] Azoulay E, Chaize M, Kentish-Barnes N. Involvement of ICU families in decisions: Fine-tuning the partnership. Annals of Intensive Care volume. 2014; 4:37. [DOI:10.1186/s13613-014-0037-5] [PMID] [PMCID]
- [12] Lynn J. Strategies to ease the burden of family caregivers. JAMA. 2014; 311(10):1021-2. [DOI:10.1001/jama.2014.1769] [PMID]
- [13] Rabiei L, Eslami AA, Abbasi M, Afzali SM, Hosseini SM, Masoudi R. Evaluating the effect of family-centered intervention program on care burden and self-efficacy of hemodialysis patient caregivers based on social cognitive theory: A randomized clinical trial study. Korean Journal of Family Medicine. 2020; 41(2):84-90. [DOI:10.4082/kjfm.18.0079] [PMID] [PMCID]
- [14] Khan TS, Hirschman KB, McHugh MD, Naylor MD. Self-efficacy of family caregivers of older adults with cognitive impairment: A concept analysis. Nursing Forum. 2021; 56(1):112-26. [DOI:10.1111/ nuf.12499] [PMID] [PMCID]
- [15] Rabei S, Ramadan S, Abdallah N. Self-efficacy and future anxiety among students of nursing and education colleges of Helwan university. Middle East Current Psychiatry. 2020; 27(1):1-5. [DOI:10.1186/s43045-020-00049-6]
- [16] Abdul Halain A, Tang LY, Chong MC, Ibrahim NA, Abdullah KL. Psychological distress among the family members of Intensive care unit (ICU) patients: A scoping review. Journal of Clinical Nursing. 2022; 31(5-6):497-507. [DOI:10.1111/jocn.15962] [PMID]
- [17] Kross EK, Engelberg RA, Gries CJ, Nielsen EL, Zatzick D, Curtis JR. ICU care associated with symptoms of depression and posttraumatic stress disorder among family members of patients who die in the ICU. Chest Journal. 2011; 139(4):795-801. [DOI:10.1378/ chest.10-0652] [PMID] [PMCID]
- [18] Hojjat-Assari S, Rassouli M, Madani M, Heydari H. Developing an integrated model of community-based palliative care into the primary health care (PHC) for terminally ill cancer patients in Iran. BMC Palliat Care. 2021;20(1):100. [DOI:10.1186/s12904-021-00795-2] [PMID] [PMCID]..
- [19] Rückholdt M, Tofler GH, Randall S, Buckley T. Coping by family members of critically ill hospitalised patients: An integrative review. International Journal of Nursing Studies. 2019; 97:40-54. [DOI:10.1016/j.ijnurstu.2019.04.016] [PMID]

- [20] Heydari H, Rahnavard Z, Ghaffari F. Exploring the position of community-based nursing in Iran: A qualitative study. International Journal of Community Based Nursing & Midwifery. 2017; 5(4):386-96. [PMID] [PMCID]
- [21] Babaei S, Abolhasani S. Family's supportive behaviors in the care of the patient admitted to the cardiac care unit: A qualitative study. Journal of Caring Science. 2020; 9(2):80-6. [DOI:10.34172/ jcs.2020.012] [PMID] [PMCID]
- [22] Davidson JE, Aslakson RA, Long AC, Puntillo KA, Kross EK, Hart J, et al., Guidelines for family-centered care in the neonatal, pediatric, and adult ICU. Critical Care Medicine. 2017; 45(1):103-28. [DOI:10.1097/CCM.00000000002169] [PMID]
- [23] Nolan MT, Hughes MT, Kub J, Terry PB, Astrow A, Thompson RE, et al. Development and validation of the family decision-making self-efficacy scale. Palliative & Supportive Care. 2009; 7(3):315-21. [DOI:10.1017/S1478951509990241] [PMID] [PMCID]
- [24] Snaith RP. The hospital anxiety and depression scale. Health and Quality of Life Outcomes. 2003; 1(1):29. [DOI:10.1186/1477-7525-1-29] [PMID] [PMCID]
- [25] Kaviani H, Seyfourian H, Sharifi V, Ebrahimkhani N. [Reliability and validity of anxiety and depression hospital scales (HADS): Iranian patients with anxiety and depression disorders (Persian)]. Tehran University Medical Journal. 2009; 67(5):379-85. [Link]
- [26] Araújo O, Lage I, Cabrita J, Teixeira L. Training informal caregivers to care for older people after stroke: A quasi-experimental study. Journal of Advanced Nursing. 2018; 74(9):2196-206. [DOI:10.1111/ jan.13714] [PMID]
- [27] Hendrix CC, Landerman R, Abernethy AP. Effects of an individualized caregiver training intervention on self-efficacy of cancer caregivers. Western Journal of Nursing Research. 2013; 35(5):590-610. [DOI:10.1177/0193945911420742] [PMID]
- [28] Kynoch K, Chang A, Coyer F, McArdle A. Developing a model of factors that influence meeting the needs of family with a relative in ICU. International Journal of Nursing Practice. 2019; 25(1):e12693. [DOI:10.1111/ijn.12693] [PMID]
- [29] National Institute for Health and Care Excellence (NICE). Rehabilitation after critical illness in adults Clinical guideline. London: National Institute for Health and Care Excellence; 2009. [Link]
- [30] Molina JA, Seow E, Heng BH, Chong WF, Ho B. Outcomes of direct and indirect medical intensive care unit admissions from the emergency department of an acute care hospital: A retrospective cohort study. BMJ Open. 2014; 4(11):e005553. [DOI:10.1136/bmjopen-2014-005553] [PMID] [PMCID]
- [31] Bibas L, Peretz-Larochelle M, Adhikari NK, Goldfarb MJ, Luk A, Englesakis M, et al. Association of surrogate decision-making interventions for critically ill adults with patient, family, and resource use outcomes: A systematic review and meta-analysis. JAMA Network Open. 2019; 2(7):e197229. [DOI:10.1001/jamanetworkopen.2019.7229] [PMID] [PMCID]
- [32] Heydari H. [Home-based palliative care: A missing link to patients' care in Iran (Persian). Journal of Hayat. 2018; 24(2):97-101. [Link]
- [33] Heydari H, Hojjat-Assari S, Almasian M, Pirjani P. Exploring health care providers' perceptions about home-based palliative care in terminally ill cancer patients. BMC Palliative Care. 2019; 18(1):66. [DOI:10.1186/s12904-019-0452-3] [PMID] [PMCID]